Author's response to reviews

Title: Feasibility of a multidimensional home-based exercise programme for the elderly with structured support given by the general practitioner's surgery: Study protocol of a single arm trial preparing an RCT [ISRCTN58562962]

Authors:

Timo Hinrichs (timo.hinrichs@rub.de)
Claudio Bucchi (claudio.bucchi@rub.de)
Michael Brach (michael.brach@uni-muenster.de)
Stefan Wilm (stefan.wilm@uni-wh.de)
Heinz G Endres (heinz.endres@rub.de)
Ina Burghaus (ina.burghaus@rub.de)
Hans-Joachim Trampisch (hans.j.trampisch@rub.de)
Petra Platen (petra.platen@rub.de)

Version: 2 Date: 29 July 2009

Author's response to reviews: see over
Feasibility of a multidimensional home-based exercise programme for the elderly with structured support given by the general practitioner's surgery: Study protocol of a single arm trial preparing an RCT [ISRCTN58562962] - Revision -

Dear Editor

We were asked to include a power calculation used to calculate the sample size of our study. Please find the following changes in the methods section our revision, subsection “Statistical analysis and decision-making on RCT preparation”:

1. Within paragraph “Criteria to decide on preparation of an RCT” we mention the use of binomial tests to analyse the criteria.
2. A new paragraph “Sample size” was added, describing power calculation and sample size simulation.

We are looking forward to further comments.

With kind regards

Michael Brach
(submitting author)